Author's response to reviews

Title: Caste-based social inequalities and childhood anemia: Results from the National Family Health Survey (NFHS) 2005-2006 in India

Authors:

Priya Vart (vart.priya@yahoo.co.uk)
Ajay Jaglan (jaglanajay8@gmail.com)
Kashif Shafique (kashif.shafique@Glasgow.ac.uk)

Version: 2 Date: 17 September 2014

Author's response to reviews:

Dear editorial staff

Following changes are made in manuscript.

1. Line and page number has been added
2. Competing interests heading is added
3. Author contributions are added.

Regards

Dr. Kashif